Author's response to reviews

Title: A double blind, randomised placebo controlled trial of topical 2% viscous lidocaine in improving oral intake in children with painful infectious mouth conditions

Authors:

Sandy M Hopper (sandy.hopper@rch.org.au)
Franz E Babl (franz.babl@rch.org.au)
Michelle McCarthy (michelle.mccarthy@rch.org.au)
Chasari Tancharoen (c.tancharoen@ugrad.unimelb.edu.au)
Katherine J Lee (katherine.lee@mcri.edu.au)
Ed Oakley (ed.oakley@southernhealth.org.au)

Version: 5 Date: 11 November 2011

Author's response to reviews: see over
Rachel Neilan
Executive Editor
BMC-series Journals
BioMed Central Floor 6,
236 Gray’s Inn Rd
London, WC1X 8HL

11 November 2011

Your Ref: MS: 4308180945462727

Dr Neilan

Please find the proof read and corrected second revision of BMC Pediatrics article entitled:

A double blind, randomised placebo controlled trial of topical 2% viscous lidocaine in improving oral intake in children with painful infectious mouth conditions

Please do not hesitate to contact me with any questions.

Yours sincerely

Ed Oakley
Director Paediatric Emergency Medicine, Southern Health Chair Paediatric Research in Emergency Departments International Collaborative (PREDICT)

e: ed.oakley@southernhealth.org.au
p: +61 3 95942707